Author’s response to reviews

Title: Psychological wellbeing and physical activity in children and adolescents with inflammatory bowel disease compared to healthy controls

Authors:

Laura Mählmann (laura.maehlmann@gmail.com)
Markus Gerber (markus.gerber@unibas.ch)
Raoul Furlano (raoul.furlano@ukbb.ch)
Corinne Légeret (corinne.legeret@bluewin.ch)
Nadeem Kalak (Nadeem.Kalak@upkbs.ch)
Edith Holsboer-Trachsler (Edith.Holsboer@upkbs.ch)
Serge Brand (Serge.Brand@upkbs.ch)

Version: 2 Date: 29 Sep 2017

Author’s response to reviews:

We thank the Editor-in-Chief, the Editorial Board, and the Reviewer Dr. Ociskova for assessing our revised manuscript and proposed changes according to the feedback by the editorial board and the reviewer.

We thank you for forwarding us the detailed comments.

Further, we are happy to know that reviewer agreed with our changes and accepted the final version of the manuscript.

For the final editorial points, please find the detailed point-by-point-response below.

Furthermore, highlights and bold letters were removed and a clean version of the manuscript will be uploaded.

Again, we thank you very much for giving us the opportunity to revise the manuscript. Of course, we are ready to work on the manuscript in case further issues are identified or our responses in this revision are judged unsatisfactorily.